Author's response to reviews

Title: The effect of changing ventilator settings on indices of ventilation inhomogeneity in small ventilated lungs

Authors:

Gerd Schmalisch (gerd.schmalisch@charite.de)
Hans Proquitte (hans.proquitte@charite.de)
Charles C Roehr (christoph.roehr@charite.de)
Roland R Wauer (roland.wauer@charite.de)

Version: 5 Date: 17 August 2006

Author's response to reviews: see over
Dear Editors,

thank you very much for the acceptance of our manuscript. Changes are performed as suggested:

- Running head was removed
- Keywords were removed.
- Section headings were revised
- Introduction was renamed 'Background'.
- Materials & Methods was renamed 'Methods'.
- The key messages were removed
- Figure legends and tables were removed at the end of the text.
- A section “Additional files” was added as suggested.

Sincerely

Gerd Schmalisch